Author's response to reviews

Title: Anti-inflammatory effects of Polygonum minus (Huds) extract in in-vitro enzyme assays and carrageenan induced paw oedema

Authors:

Annie George (annie.g@biotropicsmalaysia.com)
Sasikala Chinnappan (sasikala.c@biotropicsmalaysia.com)
Meena Chintamaneni (meenaep08@gmail.com)
Vandana Kotak C (project@ethixpharma.com)
Yogendra Choudhary (ceo@ethixpharma.com)
Thomas Kueper (thomas_kueper@arcor.de)
Ammu K Radhakrishnan (ammu_radhakrishnan@imu.edu.my)

Version: 2
Date: 21 March 2014

Author's response to reviews:

Dear Editor,

We have submitted the full research article to your journal on 12/2/2014. Based on your response dated 8/3/2014 we have amended as shown in cover letter attached herewith.

We have made the above amended based on guidelines given in the journal's home page. We hope we will receive your kind consideration in evaluating this manuscript to broaden the body of knowledge of this very unusual herb.

Thank you very much for kind attention regarding our manuscript.

Sincerely,

Ms Annie George